ORIGINAL RESEARCH



A Hospital-Based Infant Safe Sleep Intervention and Safe Sleep Practices Among Young Women: A Prospective Longitudinal Study

Sunny H. Shin^{1,2} · Changyong Choi² · Shu-Fang Shih³ · Camie A. Tomlinson² · Tiffany Kimbrough⁴

Accepted: 21 May 2023 / Published online: 12 June 2023 © The Author(s), under exclusive licence to Springer Science+Business Media, LLC, part of Springer Nature 2023

Abstract

Introduction The rates of sudden unexpected infant death (SUID) are still high in the U.S. The longitudinal effects of SUID preventive education on infant safe sleep practices are less known. The current study evaluated the effects of a comprehensive hospital-based, SUID preventive intervention on safe infant sleep practices in the first six months of life and to identify factors associated with infant sleep practices.

Methods Using a one-group pretest and multiple posttest design, the current quantitative study examined the impacts of the infant safe sleep intervention among 411 women recruited at a large, urban, university medical center. Participants were prospectively followed and completed four surveys from childbirth. Linear mixed models were used to evaluate the effects of the SUID prevention program on four sleep practice outcomes, including removing unsafe items from the sleeping environment, bed sharing, room sharing without bed sharing, and placing the infant in a supine sleep position.

Results Compared to the baseline, participants were less likely to use unsafe items (e.g., soft bedding) in infants' sleeping areas over time. However, we found that participants reported more frequent bed sharing at 3-month and 6-month follow-ups, compared to the baseline.

Conclusions Overall, maternal education and family income were positively related to healthy infant safe sleep practices. A hospital-based preventive intervention pairing an educational initiative with home-visiting services might improve safe sleep practices to remove accidental suffocation risks from the infant sleep environment.

Significance Statement

What is already known on this subject? Despite large public health efforts, rates of sudden unexpected infant death have stagnated across the U.S. in recent years. The majority of caregivers implement some of the American Academy of Pediatrics (AAP) safe sleep guidelines, such as safe sleep position. Few caregivers, however, follow other AAP recommendations such as not using any soft and loose bedding that present a potential suffocation hazard.

What this study adds? A hospital-based preventive intervention was effective in removing unsafe items from infants' sleeping area. The development of policy and practice initiatives to prevent the use of unsafe items in an infant sleeping environment should be further encouraged.

Keywords Safe Sleep Practices · Sudden Unexpected Infant Death (SUID) · Sudden Infant Death Syndrome (SIDS) · Prevention · Infant Mortality

Sunny H. Shin sshin@vcu.edu

- ³ Department of Health Administration, College of Health Professions, Virginia Commonwealth University, 900 E. Leigh St., Camie A. Tomlinson, MSW, Box 980203, Richmond, VA 23298, USA
- ⁴ Department of Pediatrics, School of Medicine, Virginia Commonwealth University, 1000 E. Broad St, Box 980264, Richmond, VA 23298, USA

¹ Department of Psychiatry, School of Medicine, Virginia Commonwealth University, 1000 Floyd Avenue, Richmond, VA 23284, USA

² School of Social Work, Virginia Commonwealth University, 1000 Floyd Avenue, Richmond, VA 23284, USA

Sudden unexpected infant death (SUID) refers to the unexpected death of a child less than one year old without a known cause of death prior to investigation, including sudden infant death syndrome (SIDS), accidental suffocation and strangulation in bed, and unknown causes of death. Although SUID rates declined during the 1990s, rates have recently stagnated across the U.S. (Bass et al., 2018; Lambert et al., 2018). In 2019, about 3,500 infants died of SUID in the U.S. (Centers for Disease Control and Prevention, 2021).

Although there are inconsistencies in defining and classifying SUID types, a majority of SUID cases involve unhealthy infant sleep practices such as bed-sharing and unsafe items in infants' sleeping areas (e.g., blankets, pillows, stuffed animals). For example, using the Centers for Disease Control and Prevention (CDC) SUID Case Registry of 4,929 infants, Parks and colleagues found that about 72% of SUID and 75% of suffocation deaths were attributable to unsafe sleeping areas (Parks et al., 2021).

There have been many efforts to reduce sleep-related infant deaths. According to a review by Ward and Balfour (2016), most studies used multifaceted interventions including one-on-one or group education, printed materials, videos, and providing resources (e.g., cribs and wearable blankets). However, little is known regarding the effectiveness of combinations of educational videos, providing appropriate sleeping materials, and using home-visiting services to reinforce the initial educational intervention. Comprehensive prevention and education strategies focusing on removal of hazardous items from infants' sleeping area, avoidance of bed-sharing, and infant sleep position might be particularly effective in preventing SUID. In addition, few longitudinal studies have investigated whether parents/caregivers maintain infant safe-sleep practices after the initial intervention (Moon et al., 2016).

Previous studies have shown that women with low socioeconomic status were less likely to adopt infant safe-sleep practices, therefore, preventive interventions focusing on this high-risk populations are particularly important. Longitudinal and cross-sectional studies have reported that maternal education and family income are strong predictors of compromised infant safe-sleep practices (Colson et al., 2013; Hauck et al., 20,008; Hirai et al., 2019). For instance, using national trend data from 1993 to 2010, Colson and colleagues (2013) found that women with household income below \$20,000) and less than high school education were 1.7 and 1.4 times more likely to bed-share with an infant, respectively. The present study examined the effects of a hospital-based safe-sleep intervention on safe-sleep practices during the first six months of infants' lives. Using a prospective longitudinal sample of women, our study also identified factors associated with infant sleep practices.

Methods

Study Design

Our study utilized the data collected from the Longitudinal Infant and Family Environment (LIFE) study, which employed a one-group pretest and multiple posttest design to decrease unsafe sleep practices and to prevent SUID. The LIFE intervention program consisted of three components: a 15-minute educational video on infant safe-sleep practices, a baby box with a firm foam mattress and a cotton fitted sheet, and a postpartum home visit. The video recommended four safe-sleep practices from the American Academy of Pediatrics (AAP), which included placing the baby on their back to sleep, room-sharing without bed-sharing, removing unsafe items, and reducing exposure to smoke and other chemicals (AAP Task Force on Sudden Infant Death Syndrome, 2016). The baby box was provided to reduce the financial burden associated with purchasing safe sleeping materials. Trained nurses conducted home visits 7 days after hospital discharge assessing both the infant and mother and providing a brief refresher course on infant safe-sleep practices. Additional details about the study design are described elsewhere (Shin et al., 2019).

Procedures

Participants were recruited from a large, urban, university medical center in the Mid-Atlantic region. The exclusion criteria included women who were under 18 years old and non-English speakers. Those who were not excluded were provided with detailed information regarding the intervention study by our team members. Informed consent procedures were conducted with all participants prior to their participation. The LIFE study collected data four times to track how safe sleep practices changed over time using selfadministered surveys. Baseline data were collected prior to intervention (i.e., prior to watching the educational video) in the hospital (Time 1). This data reflects participants' awareness of or planned behaviors regarding safe infant sleep practices at home. Follow-up data were collected during the 7-day home visit (Time 2) as well as through online surveys at 3 months (Time 3) and 6months (Time 4). Participants were asked to report on their actual practices related to infant sleep during the Time 2-4 surveys, which took about 5 to 10 min to complete. Participants received a gift card after each completed survey. All study procedures were approved by the Virginia Commonwealth University institutional review board.

Participants

A total of 895 eligible participants provided consent and completed the baseline survey. The attrition rate were 28.8% from baseline to Time 2, 39.2% from Time 2 to Time 3, and 7.8% from Time 3 to Time 4. In this study, we included participants who completed at least three surveys (45.9%; n=411). We compared the characteristics of the analytic sample and those excluded from analysis. The results showed that our study sample had a higher likelihood of having higher income and postsecondary education, being married, and being primiparous compared to those who were excluded from analysis due to attrition (all p < .05).

Measures

We assessed four infant sleep practice outcomes, including the use of unsafe items, bed-sharing, room-sharing without bed-sharing, and sleep position. Participants reported their use of various types of soft bedding (e.g., blankets, pillows) in the past 7 days and 3 months during the follow-up surveys (Time 2–4). According to the American Academy of Pediatrics (REF), bedsharing is defined as "parent(s) and infant sleeping together on any surface (bed, couch, chair).

Participants who reported any use of these soft bedding items in the past 7 days or past 3 months, were defined as using unsafe sleep items. The question regarding bedsharingasked participants how often their infant slept in the same bed with them or anyone else in the past 7 days. Similarly, the question about room-sharing without bedsharing asked whether their baby box/crib/bed was in the same room where they slept in the past 7 days when infant slept alone. Responses to both questions were coded from 0 (never) to 4 (always). Participants also answered questions about their infant's sleep position, with response options of "on side," "on stomach," or "on back." In our analysis, we classified these responses into either supine and non-supine sleep positions. The questions for the baseline survey were different, using the phrasing "how often will you/in which ways will you" instead of "in the past 7 days."

We also collected information regarding infant characteristics such as sex, birth order, preterm birth (defined as being born more than three weeks before the due date) and low birth weight (defined as birth weight <2500 g). Maternal sociodemographic background was also obtained, including race/ethnicity, age, education, marital status, and gross annual household income.

Statistical Analysis

Descriptive statistics were used to summarize the characteristics of the study sample and parents' infant safe-sleep practices during the first 6 months of life. We used linear mixed models (LMMs) to evaluate changes in sleep practices over time and to investigate the associations between maternal/child characteristics and infant safe-sleep practices. LMMs allowed us to analyze within-subject and between-subject effects in a model simultaneously and provided relatively robust results for longitudinal data with unequal time spaces and missing observations (Pusponegoro et al., 2017). For binary outcome variables such as sleep position, we used a generalized linear mixed model (GLMM) assuming a binomial logistic distribution. The time effect was included in the mean structure of the mixed model along with covariates, while a first-order autoregressive variance-covariance matrix was assumed for the error terms. Statistical significance of factors was evaluated using beta coefficients for LMMs and odds ratio for GLMMs. All analyses were conducted using IBM SPSS.

Results

On average, participants were 27.6 years old at baseline (see Table 1). The majority of sample participants (57.9%) was non-Hispanic Black, 28.2% non-Hispanic white, and about 9% were Hispanic. About 65.7% of the sample had a high school/GED or less, and over half (56.9%) of participants reported gross annual household income less than \$25,000. Furthermore, about 60.6% of the sample were married or cohabiting while the rest were never married, separated, divorced, or widowed. In terms of infants' characteristics, nearly half were female (48.2%) and firstborns (45.8%). About 17% of infants were born preterm, and 10.2% had low birth weight.

Table 2 presented the prevalence of infant sleep practices during the first six months of life. Before watching the educational video and hospital discharge, the majority of women (75.6%) reported their intention to use one or more unsafe items in their baby's sleeping areas. However, the actual use of unsafe items in sleeping areas decreased over time, with 65.2% reporting it at Time 2, 52.2% at Time 3, and 52.1% at Time 4. In contrast, while only 3.6% of women reported their intention to share a bed with their infants "often" or "always" at baseline, this percentage increased over time to6.7% at Time 2, 18.7% at Time 3, and 22.1% at Time 4. Furthermore, the frequency of reporting "often" or "always" room-sharing without bed-sharing, decreased at Time 4 (76.2%) from 92.8% at Time 2 and 90.7% at Time 3. All participants at baseline and 97.5% of the sample at Time 2 reported they planned to use or actually used the supine position with their infants. However, the percentage of women who put their babies to sleep on their back decreased from (83.4%) at Time 3 to 78.7% at Time 4. Due

 Table 1
 Sociodemographic characteristics and birth outcomes of the sample

Variable	п	%
Maternal age		
< 20	29	7.1
≥ 20	382	92.9
Maternal race/ethnicity		
African American	238	57.9
Non-Hispanic White	116	28.2
Hispanic	35	8.5
Others	22	5.4
Maternal education		
\leq high school or GED	270	65.7
> high school or GED	141	34.3
Marital status		
Married or cohabiting	248	60.6
Never married, separated, divorced, or widowed	161	39.4
Gross annual income (\$US)		
≤ \$25,000	231	56.9
\$25,000 - \$39,999	67	16.5
\$40,000 - \$49,999	33	8.2
\$50,000 - \$74,999	31	7.6
\$75,000 - \$99,999	15	3.7
> \$100,000	29	7.1
Infant sex		
Male	205	51.8
Female	191	48.2
Birth order		
First offspring	187	45.8
Second or higher offspring	221	53.8
Preterm birth		
Yes	67	16.9
No	329	83.1
Birth weight (g)		
< 2500	42	10.2
≥ 2500	369	89.8

to the highly skewed nature of the sleep position variable, it created convergence issues in conducting LMM, therefore sleep position outcomes were excluded from the multivariable LMM analyses.

To investigate factors associated with infant safe-sleep practices, we used a logistic GLMM for the unsafe items outcome and LMMs for bed-sharing and room-sharing without bed-sharing outcomes. We controlled for maternal age, race/ethnicity, educational attainment, marital status, family income, birth order, duration of pregnancy, birth weight, and infant sex (see Table 3). Women with a high school diploma/GED or less (OR = 1.89, p < .001) and first-time mothers (OR = 1.53, p < .05) were more likely to use at least one unsafe item in their infants' sleeping areas than those with higher education level and who had previously given birth, respectively. Women with higher gross annual household income were less likely to place unsafe items in

 Table 2 Prevalence of infant sleep practices in the first six months of

Variable	T1	T2	T3	T4
Unsafe items in infant sleeping	75.6%	65.2%	52.2%	52.1%
areas				
Sleep position (Supine)	100.0%	97.5%	83.4%	78.7%
Bed sharing				
Always	0.5%	1.1%	11.0%	11.6%
Often	3.2%	5.6%	7.7%	10.5%
Sometimes	22.9%	17.0%	12.2%	11.3%
Rarely	17.5%	19.0%	25.4%	20.7%
Never	56.0%	57.3%	43.6%	45.9%
Room sharing without bed sharing*				
Always	-	90.5%	85.1%	74.1%
Often	-	2.2%	5.5%	2.1%
Sometimes	-	2.5%	2.6%	7.5%
Rarely	-	1.4%	1.5%	2.4%
Never	-	3.3%	5.2%	13.9%

Note. T1 (Time 1)=Baseline assessment; T2=7-day follow-up; T3=3-month follow-up; T4=6-month follow-up

* Not measured at T1

their babies' sleeping areas (OR = 0.70, p < .001) compared to their counterparts. Our results also showed that women were less likely to use unsafe items in infants' sleeping areas over time, compared to their planned behavior before the intervention (ORs = 0.50 at Time 2, 0.30 at Time 3, 0.33 at Time 4, all p < .001).

With respect to bed-sharing, our study found that non-Hispanic African American (b=0.37, p<.01), Hispanic (b=0.81, p<.001), and other race women (b=0.46, p<.01)tended to share a bed with their infants more frequently non-Hispanic white women. After adjusting for covariates, our results showed that bed-sharing became more frequent at Time 3 (b = 0.44, p < .001), and Time 4 (b = 0.49, p < .001). In terms of room-sharing without bed-sharing, our analysis showed that Hispanic women (b=0.36, p<.05) were more likely to room-share without bed-sharing with their babies compared to non-Hispanic White women. Moreover, women with higher gross annual household income shared rooms with their babies less frequently (b=-0.18, p<.001) compared to those with lower income. Finally, our results indicated that less frequent room-sharing with infants was reported at Time 4 compared to Time 2 (b=-0.49, p < .001).

Discussion

The present study examined the effect of a comprehensive, hospital-based, infant safe-sleep intervention on changes in safe-sleep practices among a sample of women primarily from socio-economically disadvantaged and minoritized racial/ethnic backgrounds. The intervention demonstrated

Table 3 Factors associated with infant sleep practice from multivariable linear mixed models

	Unsafe items		Bed sharing		Room sharing without bed sharing	
	OR	95% CI	b	95% CI	b	95% CI
Intercept	2.65	0.60-11.65	-0.21	-0.76-0.76	3.96***	3.25-4.67
Maternal race/ethnicity						
Non-Hispanic White	Ref.		Ref.		Ref.	
African American	1.38	0.91-2.10	0.37^{**}	0.15-0.59	0.16	-0.07-0.38
Hispanic	1.61	0.83-3.12	0.81^{***}	0.46-1.15	0.36*	0.01 - 0.71
Other	1.65	0.84-3.23	0.46^{**}	0.10-0.82	0.24	-0.12-0.61
Maternal educational attainment						
> high school/GED	Ref.		Ref.		Ref.	
\leq high school/GED	1.89**	1.29-2.91	0.18	-0.06-0.42	0.08	-0.17-0.32
Maternal marital status						
Single	Ref.		Ref.		Ref.	
Married/cohabiting	1.01	0.72-1.43	-0.01	-0.19-0.17	-0.07	-0.25-0.12
Maternal age	0.99	0.96-1.03	0.01	-0.01-0.26	0.01	-0.01-0.03
Family income	0.70^{***}	0.61-0.81	-0.04	-0.12-0.03	-0.18***	-0.25 0.10
Birth order						
Second or higher offspring	Ref.		Ref.		Ref.	
First offspring	1.53*	1.10-2.10	-0.01	-0.17-0.15	-0.01	-0.18-0.16
Duration of pregnancy						
Full-term birth	Ref.		Ref.		Ref.	
Preterm birth	0.92	0.59-1.41	0.01	-0.21-0.24	-0.09	-0.32-0.13
Birth weight (kgs)	1.13	0.84-1.52	0.05	-0.10-0.20	-0.08	-0.20-0.05
Infant sex						
Male	Ref.		Ref.		Ref.	
Female	0.93	0.69-1.24	-0.03	-0.18-0.13	0.04	-0.11-0.20
Time (T)						
T1	Ref.		Ref.		-	-
T2	0.50^{***}	0.37-0.68	-0.00	-0.14-0.13	Ref.	Ref.
Т3	0.30***	0.22-0.42	0.44^{***}	0.28 - 0.60	-0.08	-0.23-0.06
T4	0.33***	0.23–0.46	0.49***	0.33-0.65	-0.49***	-0.65 0.33

Note. GED=General Educational Development; CI=Confidence interval; Ref. = Reference group of a categorical variable; T1=Baseline; T2=7-day follow-up; T3=3-month follow-up; T4=6-month follow-up. Room sharing was not measured at T1 $^{*}p < .05$. $^{**}p < .01$. $^{***}p < .001$

positive effects on reducing unsafe items in infant sleeping areas over time. Our results also indicated that maternal education and family income were negatively associated with using unsafe items in infant sleeping areas. With regard to bed-sharing, our findings suggest that there were racial/ ethnic differences, with women from minoritized racial/ethnic backgrounds being more likely to bed-share with their infants compared to non-Hispanic White women. Infant characteristics were not significantly associated with safesleep practices except for low birth order being positively related to the use of unsafe items in the sleeping area.

Our results provide evidence that our intervention was effective in reducing the use of unsafe items in the infant sleep environment over time. According to the baseline survey, most women were aware of safe-sleep practices for sleep position and location (i.e., room-sharing without bed-sharing), which was consistent with previous research (Ahlers-Schmidt et al., 2014; Mathews et al., 2016; Thach et al., 2007). However, over 75% of women were not aware of the risks of soft bedding materials in the sleeping environment before the intervention. This highlights the importance of educational programs like our intervention, to reduce the risk of SUID. Prevention strategies could be improved by educating caregivers on the danger of soft bedding and encouraging them to remove unsafe items from infant sleeping areas to promote safe sleep practices.

Our intervention program was designed based on the 2016 AAP recommendations, therefore, it is possible that our results may have been influenced by concurrent evidence-based infant safe-sleep practices recommended by pediatricians. Although we were unable to account for how much information women received through other channels

such as pediatricians, this may help to explain why our findings differ from other studies. For example, Mathews et al. (2016) conducted a randomized controlled trial among African American women and did not observe significant effects on messaging emphasizing safe-sleep practices according to the AAP for the reduction of SIDS and use of unsafe bedding items. However, messaging emphasizing safesleep practices to reduce the risk of SIDS and suffocation significantly reduced the use of soft bedding in their study. Given that the use of soft bedding remains one of the leading causes of suffocation deaths (Parks et al., 2021), future research should focus on identifying the most effective strategies to modify caregivers' behaviors and reduce the use of unsafe items in the sleeping area. One possible direction is to explore whether consistency of information shared with women by both an intervention program and their pediatricians may further decrease this practice.

In contrast to the observed reduction of unsafe items. our study found a decline in adherence to other infant safesleep practices decreased over time. Specifically, although most women reported avoiding bed-sharing with their infant throughout the study period, the percentage of women who bed-shared always or often increased after the third postpartum month. These findings are consistent with previous research indicating an increase in prone sleeping and bed-sharing during the first six months of life (Jenni et al., 2005; Lesko et al., 1998; Stremler et al., 2013). One possible explanation for the observed increase in bed-sharing over time is related to infant development. As infants grow and gain motor skills, parents may perceive that prone sleep position or bed-sharing are less risky. Additionally, our findings suggest that women reported room-sharing with their infant without bed-sharing frequently (i.e., often or always) throughout the study period (92.7% at Time 2, 90.6% at Time 3, 76.2% at Time 4). This high frequency of roomsharing may contribute to the increase in bed-sharing over time. For example, Paul et al. (2017) found that room-sharing is associated with greater odds of later bed-sharing during the night among parents of 4- and 9-month-old infants compared to parents whose infants sleep in a separate room. Although the majority of SUID occur during the first four months of life, it is still critical to adhere to safe-sleep guidelines to reduce the risk of SUID until the infant's first birthday (Trachtenberg et al., 2012).

Although our intervention included education on safesleep practices during the 7-day home visit, our study found a decline in safe-sleep practices related to sleep position, bed-sharing, and room-sharing without bed-sharing over time. These results suggest that more intensive strategies for follow-up interventions may be beneficial to maintain safesleep practices. As discussed previously, pediatricians may play a critical role in providing education and reinforcing safe-sleep practices. Therefore, future research should explore the effects of consistent messaging from a hospitalbased intervention program and pediatricians on promoting and sustaining safe-sleep practices.

It is important to contextualize our findings within the sociocultural background of our study sample, which predominantly consisted of minoritized racial/ethnic women with low income and educational attainment. Our study found that women from marginalized groups were more likely to use unsafe sleep practices over time. This finding is consistent with prior research, that reported that women with low socioeconomic status and minoritized racial/ethnic backgrounds engaged in unsafe infant sleep practices due to limited resources to create safe-sleep environments or having cultural differences in sleep practices (Colson et al., 2013; Hauck et al., 2008; Herman et al., 2015; Oden et al., 2010). It is important to acknowledge that social and cultural factors may play a significant role in shaping safesleep practices and to develop interventions that are culturally sensitive and tailored to the specific needs of these populations.

Although we used the AAP recommendation to create the educational video, the messages were not tailored to families with low income and educational background, or to be culturally-relevant to minoritized racial/ethnic groups. As these groups seemed to benefit the least from the intervention, it is important to consider whether the intervention is effective within these populations and explore culturally-relevant intervention strategies to improve safe-sleep practices. In addition, it is necessary to identify alternative safe-sleep practices that would honor cultural traditions while reducing risk of SUID (Tipene-Leach et al., 2014). Several studies have found that parents' knowledge and beliefs about comfort, safety, and convenience of certain sleep practices can influence caregiver's decisions regarding infant sleep location, position, and environments (Carlin & Moon, 2017; Gaydos et al., 2015). Thus, qualitative studies that obtain caregiver perceptions and knowledge of safe-sleep practices can reveal cultural barriers to infant safe-sleep practices and potential strategies to better engage at-risk populations to promote safe-sleep practices and health equity among minoritized racial/ethnic groups.

It is also important to consider the intersection of low income, lack of health insurance and/or healthcare access. Women who cannot receive regular healthcare or infant well-child visits may miss out on education regarding safesleep practices; therefore, it is essential to proactively provide educational interventions prior to hospital discharge after birth and through other avenues. For example, public awareness campaigns and education strategies can focus on innovative message delivery systems to reach a broader audience, particularly women from marginalized backgrounds.

Strengths and Limitations

Our study has strengths in terms of using a comprehensive approach to prevent SUID, a relatively large sample of atrisk women, and a prospective longitudinal design. However, the findings must be considered within its limitations. First, baseline measures reflect women's intentions rather than actual behaviors. Therefore, our findings from baseline to Time 2 might reflect the intention-to-action transition rather than changes in actual behaviors. Second, the lack of a control group in this non-RCT study limits the ability to draw, causal inferences about the effectiveness of the intervention. Although an RCT is the gold standard for determining effectiveness of an intervention, it is not feasible for many safe-sleep interventions (Moon et al., 2016). Third, our study excluded women who were younger than 18 years and non-English speaking which may limit the generalizability of the findings. Minors, in particular, may benefit from comprehensive safe-sleep initiatives and home-visiting services due to limited child development knowledge and resources. Moreover, cultural differences in infant sleep practices were not examined although cultural influences may play a role in safe sleep. Next, the current study was not able to include all risk and protective factors for infant safe sleep such as breastfeeding or birth experiences. For example, while the birthing person's birth experience, particularly a traumatic birth, may have contributed to their awareness and engagement with our intervention, the current study did not examine such variables. Further research would be needed to fully explore the roles of birth experiences in the development and maintenance of infant safe sleep practices. Finally, the study sample was limited to women from the Mid-Atlantic region of the U.S. and those included in the analysis differed from those excluded from analysis in terms of, income, education levels, marital status, and parity. Therefore, the interpretation of findings may be specific to the characteristics of the study sample.

In conclusion, our results suggest that a hospital-based preventive intervention incorporating education and homevisiting services can effectively improve safe-sleep practices by reducing accidental suffocation risks in the first six months of life through removing unsafe items from the sleeping areas. However, the intervention did not demonstrate the effectiveness in a reduction in bed-sharing. Furthermore, we still continue to see that underlying sociocultural factors influencing parents' decisions to continue to bed-share. Therefore, it is imperative that future research focuses on specific target groups and develop culturally appropriate intervention strategies that effectively reduce the use of unsafe sleep practices to reduce the rate of SUIDs and promote health equity across populations.

Acknowledgements This research was funded by the Virginia Department of Social Services (FAM 17–084 to SHS).

Author contributions All authors whose names appear on the submission: (a) made substantial contributions to the conception or design of the work; or the acquisition, analysis, or interpretation of data; or the creation of new software used in the work; (b) drafted the work or revised it critically for important intellectual content; (c) approved the version to be published; and (d) agree to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved.

Data Availability The datasets generated during and/or analyzed during the current study are available from the corresponding author on reasonable request.

Code Availability Not applicable.

Declarations

Conflicts of interest The authors have no conflicts of interest and/or competing interests to declare.

Ethics Approval This study was approved by the Virginia Commonwealth University IRB.

Consent to Participate Informed consent procedures were established and approved by the University IRB. All participants provided written consent prior to data collection.

Consent for publication The consent procedures/form participants signed gave the study investigators permission to analyze and publish results in their aggregate form.

References

- AAP Task Force on Sudden Infant Death Syndrome. (2016). SIDS and other sleep-related infant deaths: Updated 2016 recommendations for a safe infant sleeping environment. *Pediatrics*, *138*(5), https://doi.org/10.1542/peds.2016-2938. Article e20162938.
- Ahlers-Schmidt, C. R., Kuhlmann, S., Kuhlmann, Z., Schunn, C., & Rosell, J. (2014). To improve safe-sleep practices, more emphasis should be placed on removing unsafe items from the crib. *Clinical Pediatrics*, 53(13), 1285–1287. https://doi. org/10.1177/0009922813518964.
- Bass, J. L., Gartley, T., Lyczkowski, D. A., & Kleinman, R. (2018). Trends in the incidence of sudden unexpected infant death in the newborn: 1995–2014. *The Journal of Pediatrics*, 196, 104–108. https://doi.org/10.1016/j.jpeds.2017.12.045.
- Carlin, R. F., & Moon, R. Y. (2017). Risk factors, protective factors, and current recommendations to reduce sudden infant death syndrome: A review. JAMA Pediatrics, 171–180(2), 175. https://doi. org/10.1001/jamapediatrics.2016.3345
- Centers for Disease Control and Prevention (2021, April 29). Data and statistics for SIDS and SUID. Data and statistics for SIDS and SUID. https://www.cdc.gov/sids/data.htm
- Colson, E. R., Willinger, M., Rybin, D., Heeren, T., Smith, L. A., Lister, G., & Corwin, M. J. (2013). Trends and factors associated

🙆 Springer

with infant bed sharing, 1993–2010: The National infant sleep position study. *JAMA Pediatrics*, *167*(11), 1032–1037. https://doi.org/10.1001/jamapediatrics.2013.2560.

- Gaydos, L. M., Blake, S. C., Gazmararian, J. A., Woodruff, W., Thompson, W. W., & Dalmida, S. G. (2015). Revisiting safe sleep recommendations for african-american infants: Why current counseling is insufficient. *Maternal and Child Health Journal*, 19(3), 496–503. https://doi.org/10.1007/s10995-014-1530-z.
- Hauck, F. R., Signore, C., Fein, S. B., & Raju, T. N. K. (2008). Infant sleeping arrangements and practices during the first year of life. *Pediatrics*, 122(2), 113–120. https://doi.org/10.1542/ peds.2008-13150.
- Herman, S., Adkins, M., & Moon, R. Y. (2015). Knowledge and beliefs of african-american and american indian parents and supporters about infant safe sleep. *Journal of Community Health*, 40(1), 12–19. https://doi.org/10.1007/s10900-014-9886-y.
- Hirai, A. H., Kortsmit, K., Kaplan, L., Reiney, E., Warner, L., Parks, S. E., Perkins, M., Koso-Thomas, M., D'Angelo, D. V., & Shapiro-Mendoza, C. K. (2019). Prevalence and factors associated with safe infant sleep practices. *Pediatrics*, 144(5), https://doi. org/10.1542/peds.2019-1286. Article e20191286.
- Jenni, O. G., Fuhrer, H. Z., Iglowstein, I., Molinari, L., & Largo, R. H. (2005). A longitudinal study of bed sharing and sleep problems among swiss children in the first 10 years of life. *Pediatrics*, *115*(Supplement 1), 233–240. https://doi.org/10.1542/ peds.2004-0815E.
- Lambert, A. B. E., Parks, S. E., & Shapiro-Mendoza, C. K. (2018). National and state trends in sudden unexpected infant death: 1990–2015. *Pediatrics*, 141(3), 1–9. https://doi.org/10.1542/ peds.2017-3519.
- Lesko, S. M., Corwin, M. J., Vezina, R. M., Hunt, C. E., Mandell, F., McClain, M., Heeren, T., & Mitchell, A. A. (1998). Changes in sleep position during infancy: A prospective longitudinal assessment. *Journal Of The American Medical Association*, 280(4), 336–340. https://doi.org/10.1001/jama.280.4.336.
- Mathews, A., Joyner, B. L., Oden, R. P., He, J., McCarter, R., & Moon, R. Y. (2016). Messaging affects the behavior of african american parents with regards to soft bedding in the infant sleep environment: A randomized controlled trial. *The Journal of Pediatrics*, 175, 79–85e2. https://doi.org/10.1016/j.jpeds.2016.05.004.
- Moon, R. Y., Hauck, F. R., & Colson, E. R. (2016). Safe infant sleep interventions: What is the evidence for successful behavior change? *Current Pediatric Reviews*, 12(1), 67–75. https://doi.org /10.2174/1573396311666151026110148.
- Oden, R. P., Joyner, B. L., Ajao, T. I., & Moon, R. Y. (2010). Factors influencing african american mothers' decisions about sleep position: A qualitative study. *Journal of the National Medical Association*, 102(10), 870–880. https://doi.org/10.1016/ S0027-9684(15)30705-7.

- Parks, S. E., Lambert, A. B. E., Hauck, F. R., Cottengim, C. R., Faulkner, M., & Shapiro-Mendoza, C. K. (2021). Explaining sudden unexpected infant deaths, 2011–2017. *Pediatrics*, 147(5), https://doi.org/10.1542/peds.2020-035873. Article e2020035873.
- Paul, I. M., Hohman, E. E., Loken, E., Savage, J. S., Anzman-Frasca, S., Carper, P., Marini, M. E., & Birch, L. L. (2017). Mother-infant room-sharing and sleep outcomes in the INSIGHT study. *Pediatrics*, 140(1), https://doi.org/10.1542/peds.2017-0122. Article e20170122.
- Pusponegoro, N. H., Notodiputro, K. A., & Sartono, B. (2017). Linear mixed model for analyzing longitudinal data: A simulation study of children's growth differences. *Procedia Computer Science*, *116*, 284–291. https://doi.org/10.1016/j.procs.2017.10.071.
- Salm Ward, T. C., & Balfour, G. M. (2016). Infant safe sleep interventions, 1990–2015: A review. *Journal of Community Health*, 41(1), 180–196. https://doi.org/10.1007/s10900-015-0060-y.
- Shin, S. H., Jiskrova, K., Kimbrough, G., Trowbridge, T., Lee, K., E., & Ayers, C. E. (2019). Impact of a comprehensive hospital-based program to remove unsafe items from infant's sleeping area: A prospective longitudinal study. *Clinical Pediatrics*, 58(13), 1440– 1443. https://doi.org/10.1177/0009922819850479.
- Stremler, R., Hodnett, E., Kenton, L., Lee, K., Macfarlane, J., Weiss, S., Weston, J., & Willan, A. (2013). Infant sleep location: Bed sharing, room sharing and solitary sleeping at 6 and 12 weeks postpartum. *The Open Sleep Journal*, 6(1), 77–86. https://doi.org /10.2174/1874620901306010077.
- Thach, B. T., Rutherford, G. W., & Harris, K. (2007). Deaths and injuries attributed to infant crib bumper pads. *The Journal* of *Pediatrics*, 151(3), 271–274e3. https://doi.org/10.1016/j. jpeds.2007.04.028.
- Tipene-Leach, D., Baddock, S., Williams, S., Jones, R., Tangiora, A., Abel, S., & Taylor, B. (2014). Methodology and recruitment for a randomised controlled trial to evaluate the safety of *wahakura* for infant bedsharing. *BMC Pediatrics*, 14, 240. https://doi. org/10.1186/1471-2431-14-240.
- Trachtenberg, F. L., Haas, E. A., Kinney, H. C., Stanley, C., & Krous, H. F. (2012). Risk factor changes for sudden infant death syndrome after initiation of back-to-sleep campaign. *Pediatrics*, *129*(4), 630–638. https://doi.org/10.1542/peds.2011-1419.

Publisher's Note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Springer Nature or its licensor (e.g. a society or other partner) holds exclusive rights to this article under a publishing agreement with the author(s) or other rightsholder(s); author self-archiving of the accepted manuscript version of this article is solely governed by the terms of such publishing agreement and applicable law.